Author’s response to reviews

Title: Clinical and Biochemical Factors to Predict Biochemical Adrenal Insufficiency in Hospitalized Patients with Indeterminate Cortisol Levels: a Retrospective Study

Authors:

worapaka manosroi (worapaka.m@gmail.com)
Natapong Kosachunhanan (natapong.k@cmu.ac.th)
Pichitchai Atthakomol (p.attakomol@gmail.com)

Version: 2 Date: 11 Feb 2020

Author’s response to reviews:

Response to Editors’ Comments:

1. As many abbreviations are used throughout the paper, we ask that you include a list of abbreviations (before the declarations section) for reference and ease of reading. All abbreviations should still be defined in the text at first use.
   All abbreviations have been defined as requested.

2. Please clarify whether any administrative permissions were required to access the raw data. If so please state who granted permission in the “Ethics approval and consent to participate” section.
   There is no administrative permissions to access the raw data.

3. Please remove the funding information from the Acknowledgements and include it in the Funding section instead. If you have no further acknowledgements please put “Not Applicable” in the Acknowledgements section.
   The funding has been removed from the acknowledgements section.

4. Please remove any additional files you do not want published alongside your manuscript, including cover/response letters.
   Additional files have been removed.

5. Please put your responses to the reviewers'/editors’ comments in the Response to Reviewers box in Editorial Manager. Please do not upload a separate letter. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Additional files should remain uploaded as separate files. Please ensure that all figures, tables and additional-supplementary files are cited within the text.
   The manuscript has been cleaned and uploaded per your suggestion.